Author’s response to reviews

Title: Vitamin D and tuberculosis: a multicenter study in children.

Authors:

Elisabetta Venturini (elisabetta-venturini@virgilio.it)
Ludovica Facchini (facchini.ludovica@gmail.com)
Nuria Martinez-Alier (nuria.martinez-alier@gstt.nhs.uk)
Vas Novelli (Vas.Novelli@gosh.nhs.uk)
Luisa Galli (luisa.galli@unifi.it)
Maurizio de Martino (maurizio.demartino@unifi.it)
Elena Chiappini (elena.chiappini@unifi.it)

Version: 3
Date: 26 June 2014

Author’s response to reviews:

Florence, June 27, 2014
Dear Editor,

We would like to hereby ask you to consider the manuscript “Vitamin D and tuberculosis: a multicenter study in children” for the publication in BMC Infectious Diseases. We accept responsibility for the scientific integrity of the work described in this manuscript. It is not under consideration for publication elsewhere and its submission for publication has been approved by all of the authors. There is no commercial association that might pose a conflict of interest. Any copyrighted material from other source was reproduced in our manuscript. These data were not presented before and there are not closely related manuscripts submitted for simultaneous consideration to the same or another journal.

I am looking forward to receiving your replay.

Yours sincerely,

Prof. Maurizio de Martino, MD

Full Professor of Pediatrics,
Head, Department of Pediatrics, University of Florence
Anna Meyer Children’s Hospital
Viale Pieraccini, 24, I-50132 Florence, Italy
phone: 0039 055 5662494
fax: 0039 055 570380
e-mail: maurizio.demartino@unifi.it